Author's response to reviews

Title: Longitudinal study of leptin levels in chronic hemodialysis patients

Authors:

Ilia Beberashvili (iliab@asaf.health.gov.il)
Inna Sinuani (sinuanii@asaf.health.gov.il)
Ada Azar (liada@bezeqint.net)
Hila Yasur (iliabeber@yahoo.com)
Leonid Feldman (leonidf@asaf.health.gov.il)
Zhan Averbukh (AVERBUKHZH@asaf.health.gov.il)
Joshua Weissgarten (weissgarten@asaf.health.gov.il)

Version: 7 Date: 10 June 2011

Author's response to reviews:

To: Nehme Gabriel, M.D.
Editor-in-Chief, Nutrition Journal.

Subject: Revision - Nutrition Journal -1355444765525489.

Dear Editor,

Thank you for your letter concerning our manuscript # 1355444765525489 entitled “Longitudinal study of leptin levels in chronic hemodialysis patients”.

I’d like to add the word “participants” (that is missing due to technical reason m/p) in 1-st line of page 16 of the manuscript.

I have not any other corrections.

Sincerely yours,
Ilia Beberashvili, MD.
Nephrology Division, Assaf Harofeh Medical Center